Author's response to reviews

Title: Inhibition of Radiation Induced Migration of Human Head and Neck Squamous Cell Carcinoma by Blocking EGF Receptor Pathways

Authors:

Anja C Pickhard (a.pickhard@lrz.tum.de)
Johanna Margraf (Tahiti@web.de)
Andreas Knopf (a.knopf@lrz.tum.de)
Thomas Stark (t.stark@lrz.tum.de)
Guido Piontek (piontek@lrz.tum.de)
Carolin Beck (carolin.beck@mytum.de)
Anne-Laure Boulesteix (boulesteix@ibe.med.uni-muenchen.de)
Elias Q Scherer (s.e.scherer@lrz.tu-muenchen.de)
Steffi Pigorsch (s.pigorsch@lrz.tum.de)
Jürgen Schlegel (juergen.schlegel@lrz.tu-muenchen.de)
Wolfgang Arnold (w.arnold@lrz.tum.de)
Rudolf Reiter (rudolf.reiter@uniklinik-ulm.de)

Version: 8 Date: 5 August 2011

Author's response to reviews: see over
Point-by-point response

Editorial request:

Competing interests - Please include a 'Competing interests' section between the Conclusions and Authors' contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'.

A 'Competing interests' section was included.

Authors' contributions - Please include an Authors' contributions section before the Acknowledgements and Reference list.

The Authors' contributions section was included.